Author's response to reviews

**Title:** A Machine-independent Method to Have Active Removal of 5,000 Centistokes Silicone Oil Using Plastic Infusion Tube and 23-gauge Microcannulas

**Authors:**

- Zhaotian Zhang M.D. (zhangzhaotian@126.com)
- Yantao Wei M.D., Ph.D. (weiyantao@126.com)
- Xintong Jiang M.D., Ph.D. (jiangtong4@163.com)
- Suo Qiu M.D. (lucy@163.com)
- Shaochong Zhang M.D. (zhshaochong@outlook.com)

**Version:** 2  **Date:** 3 June 2015

**Author's response to reviews:**

Appropriate revision has been added to the current manuscript, including ethics statement and the clarification that all the patients have had signed informed consent for the surgery.